Author's response to reviews

Title: Cancer therapy and cardiotoxicity: The need of serial Doppler echocardiography

Authors:

Maurizio Galderisi (mgalderi@unina.it)
Francesco Marra (fra-marra@libero.it)
Oreste de Divitiis (devitiis@unina.it)

Version: 3 Date: 24 August 2006

Author's response to reviews:

In the new draft you will find:
1. declaration of "no competing interest" before the reference list
2. affiliation written in full, providing department, institute, city and country.
3. each author linked to their affiliation